Author’s response to reviews

Title: Case report: two novel VPS13B mutations in a Chinese family with Cohen syndrome and hyperlinear palms

Authors:
Sha Zhao (zhaosha666@163.com)
Zhenqing Luo (2938610199@qq.com)
Xiao Zhenghui (xiaozh888@126.com)
Liping Li (13974871993@163.com)
Rui Zhao (ruizhao@aliyun.com)
Yongjia Yang (yongjia727@aliyun.com)
Yan Zhong (zhongyan@163.com)

Version: 3 Date: 06 Nov 2019

Author’s response to reviews:

MGTC-D-19-00275R1
Case report: Two novel VPS13B mutations in a Chinese family with Cohen syndrome and hyperlinear palms
Sha Zhao; Zhenqing Luo; Xiao Zhenghui; Liping Li; Rui Zhao; Yongjia Yang; Yan Zhong
BMC Medical Genetics

Dear Editor,

Thanks again for your further procession of our manuscript. As the reception of your positive decision, we would like to express appreciation, and we spared no effort to revise the paper that is according to the comments which put forward by Editor.
Yours sincerely,

Yongjia Yang, M.D.
The Laboratory of Genetics and Metabolism
Hunan Children’s Hospital
86 Ziyuan Road, Changsha, Hunan 410007, China
Tel: +86 731 85600981
Fax: +86 731 85356666
Email: yongjia727@aliyun.com
Assistant Editor Comments:

1. Author Change Form
-- We note that the name of one author [ZX] has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.
ANS: We have downloaded the form and have filled it step by step. In filling out the form, we carefully considered the suggestions put forward by the editor. The table is sent through email.

2. Figures and Tables
-- Please ensure that all figures, tables and supplementary/additional files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
ANS: After checking, we made sure that all figures, tables and supplementary/additional files are cited within the text.

3. Authors’ Contributions
-- We have noted that author [Rui Zhao, RZ] is missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section.
Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship
-- Secondly, the contributions of author [LL, Liping Li] do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.
ANS: The previous-author contribution section has negligence. Then we have revised the section for the paper as follows:
YJY, RZ and YZ designed the study, monitored the family investigation, collected clinic data, conducted experiments, and revised the manuscript. YJY, ZQL, ZHX, LLP and SZ contributed to exome sequencing, PCR experiments, Sanger sequencing, and drafting the paper. SZ and YJY analyzed the clinical data. YJY, LLP, and SZ contributed to sample collection and genetic counselling. All authors read and approved the final manuscript.

4. Funding
-- In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
ANS: We have added the following contents to this section: The funding body had no influence in the in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

5. Formatting
--Please rename your ‘Discussion and Conclusion’ section in the main body of your manuscript to ‘Discussion and Conclusions’.
--Please ensure that your Declarations are listed beneath the ‘Declarations’ heading at the end of your manuscript in a congruent order, taking care to remove the ‘Not applicable’ statement from beneath this section heading.
--Please remove your ‘Acknowledgements’ from the main body of the manuscript, ensuring that they are included under the ‘Declarations’ section heading instead.
ANS: All the above issues have been appropriately addressed (that shown in the revised edition of the manuscript).